Author's response to reviews

Title: Human papillomavirus infections in women seeking cervical Papanicolaou cytology of Durango, Mexico: prevalence and genotypes.

Authors:

Luis Francisco Sanchez-Anguiano (lfsanguiano@hotmail.com)
Cosme Alvarado-Esquivel (alvaradocosme@yahoo.com)
Miguel Arturo Reyes-Romero (lfsanguiano@hotmail.com)
Margarita Carrera-Rodriguez (lfsanguiano@hotmail.com)

Version: 5 Date: 14 December 2005

Author’s response to reviews:


Dear Editor,

Attached please find a new version of our manuscript that had been modified according to the BMC Infectious Diseases format. We have performed the following modifications:

1. We have added an hyphen between the family name of Dr. Luis Francisco Sanchez-Anguiano.
2. We have modified Figure 1 according to the BMC instructions. We were unable to perform the modification in Microsoft Word, therefore we have change the figure to Microsoft PowerPoint. In addition, we have placed the figure legend at the end of the main manuscript just below the reference section.
3. With regard to the competing interests section, we have written "The authors declare that they have no competing interests".

Yours faithfully,

Dr. Luis Francisco Sanchez Anguiano.
Facultad de Medicina
Avenida Universidad y Fanny Anitua
34000 Durango, Dgo. Mexico.
Tel.: 0052 618 8 128009
Fax: 0052 618 8 130527
Email: lfsanguiano@hotmail.com